\documentclass[12pt]{amsart}
\usepackage{amssymb, amsmath}

\begin{document}
\begin{center}
{\Huge 18.100C Homework 4} 
\\
Due online by noon on Tuesday, October 13
\end{center}

\medskip

While you review for next Thursday's exam, formulate \emph{precise} questions related to the course content so far.  In addition to making your questions \emph{clear and precise}, you should try to make them as interesting and relevant to the course as possible.  We will address (some of) these questions in recitation.

If you feel that you have a sufficient grasp of the course material, ask interesting questions that extend beyond the boundary of what you've learned so far.

Submit your questions to Stellar in .tex and .pdf format, and bring a copy to class with you on Tuesday (or at least make sure you know what they are).

\end{document}
